Author's response to reviews

Title: Delayed diagnosis of coeliac disease increases cancer risk

Authors:

Marco Silano (marco.silano@iss.it)
Umberto Volta (uvolta@orsola-malpighi.med.unibo.it)
Anna M Mechia (mbpsegr@iss.it)
Mariarita Dessi (mariarita.dessi@ptvonline.it)
Rita Di Benedetto (ritadib@iss.it)
Massimo De Vincenzi (massimo.devincenzi@iss.it)

Version: 13 Date: 8 March 2007

Author's response to reviews:

Dear Editor,
please find enclosed the revised version of the paper. In the Methods section, the approval number by Istituto Superiore di Sanita has been added, along with the statement "all the patients involved in the study signed the authorization for the handling of the personal data and that the data were collected and stored according to national regulation" as you have suggested.

Thank you

Best Regards,

Marco Silano

Division of Food Science, Human Nutrition and Health

Istituto Superiore di Sanita

Viale Regina Elena 299

00161 Roma ITALY

e.mail: marco.silano@iss.it

phone: 0039-06-49902589

fax: 0039-06-49387149